

Author's response to reviews

Title: Implementing the WHO integrated tool to assess quality of care for mothers, newborns and children: results and lessons from five districts in Malawi

Authors:

Helen Smith (helen.smith@lstmed.ac.uk)

Atnafu Getachew Asfaw (agasfaw@unicef.org)

Kyaw Myint Aung (kaung@unicef.org)

Lastone Chikoti (chikotilas@yahoo.com)

Florence Mgawadere (Florence.Mgawadere@lstmed.ac.uk)

Luigi D'Aquino (Luigi.DAquino@lstmed.ac.uk)

Nynke van den Broek (Nynke.vandenBroek@lstmed.ac.uk)

Version: 1 Date: 23 Mar 2017

Author's response to reviews:

"The authors' response letter has been included as a supplementary file"